THE DANISH INSTITUTE FOR HUMAN RIGHTS

HUMAN RIGHTS DUE DILIGENCE LAWS: KEY CONSIDERATIONS

BRIEFING ON CIVIL LIABILITY FOR DUE DILIGENCE FAILURES

OCTOBER 2021

HUMAN RIGHTS DUE DILIGENCE LAWS: KEY CONSIDERATIONS

Authors: Gabrielle Holly and Claire Methven O'Brien

e-ISBN: 978-87-7570-027-1

Layout: Semin Alekic

© 2021 The Danish Institute for Human Rights Wilders Plads 8K DK-1403 Copenhagen K Phone +45 3269 8888 www.humanrights.dk

Provided such reproduction is for non-commercial use, this publication, or parts of it, may be reproduced if author and source are quoted.

At the Danish Institute for Human Rights we aim to make our publications as accessible as possible. We use large font size, short (hyphen-free) lines, left-aligned text and strong contrast for maximum legibility. For further information about accessibility please click www.humanrights.dk/accessibility

CONTENTS

1.	INTR	ODUCTION	5
2.	CIVIL	LIABILITY AND INTERNATIONAL HUMAN RIGHTS LAW	7
3.	CORF	PORATE LIABILITY: CAUSES OF ACTION	8
4.	UNDE	ERSTANDING DUE DILIGENCE	10
	4.1.	Defining "due diligence"	10
	4.2.	Scope of human rights due diligence: key parameters	14
5.	APPL	RAL ELEMENTS OF TORT LIABILITY AND THEIR ICATION TO CLAIMS LINKED TO HUMAN RIGHTS SES BY COMPANIES	15
	5.1.	Actionable harm	15
	5.2.	Duty of care	16
	5.3.	Standard of care	17
	5.4.	Causation	18
	5.5.	Foreseeability	19
6.	FURT	HER ISSUES	21
	6.1.	Strict liability	21
	6.2.	Liability across the supply or value chain	21
	6.3.	Due diligence defence and "safe harbour"	22
7.	CONC	LUSION	25
END	OTES		26

CIVIL LIABILITY FOR DUE DILIGENCE FAILURES

This briefing by the Danish Institute for Human Rights (the **Institute**) considers civil liability for harms linked to business activities and due diligence failures – a key consideration in the design of mandatory corporate human rights due diligence laws. This briefing is intended to support those designing, advocating for or analysing supranational or national human rights due diligence laws, whether in government, business, civil society, the media or national human rights institutions.

1. INTRODUCTION

The UN Guiding Principles on Business and Human Rights (**UNGPs**)¹ provide that all businesses have a responsibility to respect internationally-recognised human rights. This responsibility applies regardless of a business' size, sector, ownership or country of operation. It also applies independently of whether governments in a business' home or host state fulfil their duties under international human rights law to protect human rights against business-related abuses.²

The UNGPs call for all businesses to undertake human rights due diligence to operationalise their responsibility to respect human rights. States should adopt legislative and other regulatory measures "to prevent, investigate, punish and redress" business-related human rights abuses.³ Such measures are important because, to date, business adherence to the corporate respect for human rights has been inadequate.⁴

Aiming to increase corporate respect for human rights, some states have enacted new laws to require or encourage businesses to undertake human rights due diligence. Others are considering this.⁵ Mandatory due diligence laws are not explicitly required by the UNGPs or other human instruments.⁶ Nevertheless, such laws have a potential role to play as one component of a state's business and human rights regulatory framework.⁷

Corporate human rights due diligence laws may take different forms. One feature of such laws may be rules allowing civil proceedings to be brought against companies for "harm" to human rights linked to their due diligence failures. For example, France's Duty of Vigilance Law (the **French Law**) establishes obligations for large companies in France to conduct human rights due diligence, and allows such companies to be sued for harms caused by failures to publish and implement a "Vigilance Plan" in accordance with the requirements of the law. In other countries, provisions to allow new civil claims have not been included in due diligence laws.⁸ To inform public debate and support policy development, this briefing highlights key issues relating to the role of civil liability in mandatory human rights due diligence laws.

Section 2 situates the discussion of civil liability for corporate conduct in its wider international human rights law context.

Section 3 identifies existing types of liability (or "causes of action") for corporate human rights abuses, including under tort, administrative, criminal and contract laws, which new due diligence laws should take into account.

Section 3 focuses on how corporate "human rights due diligence" may be defined in due diligence laws and sketches implications of different definitions of due diligence for civil liability.

Section 4 examines the general elements of tort claims. It then outlines how these elements may apply in claims linked to human rights abuses caused by corporate due diligence failures, as provided for by due diligence laws.

Section 5 considers other issues pertaining to the design of a civil liability mechanism, such as the inclusion of strict liability offences, extension of liability across the supply chain and the availability of due diligence defences or "safe harbour" provisions.

Section 6 summarises conclusions of the briefing's analysis for legislators who are considering including civil liability provisions in mandatory corporate human rights due diligence laws and other stakeholders.

A table summarising the key features of recently enacted or proposed due diligence laws can be found at page 26 of this briefing.

2. CIVIL LIABILITY AND INTERNATIONAL HUMAN RIGHTS LAW

Discussions of whether and how to include civil liability in mandatory corporate human rights due diligence laws can quickly become complex and technical. Before embarking on detailed discussions, it is therefore worth reflecting on the role of civil liability in the context of wider human rights laws.

One important starting point is the right to an effective remedy for a violation of human rights.⁹ This right is widely recognised by international human rights treaties.¹⁰ It entails an obligation to bring to justice perpetrators, and to provide appropriate reparation to victims. While domestic legal remedies should always be available, these may vary in their details as between national legal systems, while the nature of the right in question is also a factor. Generally, the right to remedy is viewed as having both substantive and procedural dimensions, the latter entitling victims to an adequate, effective and timely investigation. Notably, however, the right to remedy is not free-standing, but presumes a violation of another protected human right has taken place, or that such a violation is at least arguable.

When it comes to non-state actors, including corporations, states may have a "positive obligation" to regulate their activities to prevent harms that would interfere with the enjoyment of human rights that the state in question is obliged to protect. Effective deterrence of third-party abuses by the state may require, depending on circumstances, the criminalisation of private actors' conduct, the adoption of other legislation or policies, or the deployment of operational measures in the case of known threats, for example.

In addition, individuals under most human rights instruments have a right to a fair and public hearing of their civil rights and obligations, within a reasonable time by an independent and impartial tribunal established by law (the civil aspect of the right to a fair trial). This can apply, even where the harm complained of in civil proceedings does not amount to violation of another human right.

Tort or other civil proceedings may, depending on the circumstances, provide an opportunity for domestic authorities both to deal with the substance of a human rights complaint and to grant appropriate relief. On the other hand, international human rights jurisprudence does not currently demand that states establish due diligence laws, or civil liability provisions within them. States have a margin of discretion in how to comply with their obligations and the extent and content of duties owed by states to victims beyond their territorial jurisdiction remains under discussion.¹¹

3. CORPORATE LIABILITY: CAUSES OF ACTION

In most jurisdictions, businesses are exposed to a range of forms of legal liability that contribute to remediating human rights abuses or which have potential to do so. Mandatory due diligence laws, and any provision for civil liability linked to due diligence under them, should take account of any such existing avenues of redress. The enactment of new due diligence laws may also influence how courts or other bodies interpret and enforce established causes of action against companies. In most jurisdictions, proceedings against companies should already find a basis in the following:

Tort: Businesses may be liable for certain forms of harm to individuals under the law of tort (sometimes referred to as the law of non-contractual obligations).¹² A breach of human rights as such is not usually a form of harm recognised in national tort laws. However, an abuse of human rights may also give rise to a tort claim where the abuse causes physical injury or damage to property, for instance.

Framing human rights abuses as tort claims may present a valuable means of providing redress and accountability for rights-holders. The availability of tort claims may also contribute to fulfilling the state's duty to ensure access to effective remedy under human rights treaties. On the other hand, tort claims may not fully capture or offer the best or an appropriate remedy for the wrong done to an individual when their human rights are abused. Tort-based remedies do not usually seek to identify root causes or provide a mechanism to resolve patterns of abuses beyond the case in question.

Generally, for a claim in tort to succeed, the following conditions must be met: the type of harm caused to the claimant must be actionable; the defendant must have owed a duty of care to the injured party; an action or omission by the defendant caused must have caused harm; the harm must have been foreseeable by the defendant; the defendant's conduct must have fallen below the required standard of care it owed to the injured party, according to the law; and a court must determine that imposing liability on the defendant in relation to the harm sustained by the victim is reasonable in all the circumstances. How a company's performance of due diligence may influence their liability under general rules of tort is discussed further in Section 5 below.

Statutory liability: In most states, businesses can be liable for breaches of specific statutory duties imposed on them, for instance, by national laws on health and safety, environment, data protection or privacy, and discrimination.¹³ When a due diligence law is passed, whether a company performs due diligence as required under it may become a relevant factor in evaluating its fulfilment of such statutory duties. Vice versa, whether companies have complied with specific statutory duties they are subject to under other laws may influence courts or regulators in evaluating a company's fulfilment of a due diligence duty, or civil liability linked to due diligence failures.

Criminal liability: Some jurisdictions permit companies, like real persons, to be held liable in criminal law. The precise legal rules around this are complex and vary across jurisdictions. In some cases, corporate criminal liability is linked to the actions or intentions of senior management¹⁴ or, alternatively, to failures of "corporate culture".¹⁵ A company may also be vicariously criminally liable for the acts of employees and agents.¹⁶

Breach of contract: This may be relevant where a victim and defendant company are parties to a contract. Failing to pay wages or maintain adequate health and safety standards, for example, might be construed as breaches of an employment contract. Company officers may also be held liable via this route.¹⁷ By virtue of the principle of privity, however, contracts do not usually create rights or duties for persons who are not party to them.¹⁸ In practice this has limited the relevance of breach of contract as a route to remedy for corporate human rights abuses beyond the employment context.¹⁹

Consumer protection: A specific type of statutory liability, consumer protection laws may prohibit false advertising, misleading and deceptive conduct by companies, and the production, sale or distribution of dangerous, fake or defective products. Such provisions can sometimes be relied on where companies claim to abide by human rights standards but fail to do so in practice²⁰ or, for example, where a failure to disclose the existence of a human rights harm in the supply chain could be taken to mislead consumers in breach of the relevant law.²¹

KEY POINTS:

- Businesses may be held liable for human rights abuses through a range of legal mechanisms
- Causes of action vary across jurisdictions but usually include tort, statutory claims, corporate criminal liability, breach of contract and actions under consumer laws
- Human rights due diligence laws should, in their design, take account of existing liability mechanisms
- Whether or not they provide for civil liability themselves, human rights due diligence laws will be likely to influence, and be influenced by, other types of corporate liability
- Tort claims can be a valuable means of providing redress and accountability to rights-holders but may not be available or be an appropriate or sufficient form of remedy for all types of corporate human rights abuse

4. UNDERSTANDING DUE DILIGENCE

4.1. DEFINING "DUE DILIGENCE"

The phrase "due diligence" has a number of different established meanings. For clarity and certainty, these different senses of "due diligence" need to be distinguished during discussions and drafting of due diligence laws. The term "due diligence" must also be carefully defined in legislation and subsequent guidance for businesses, regulators or other actors.

- **Corporate human rights due diligence under the UNGPs:** In the UNGPs, human rights due diligence is a term used to describe a cyclical (or 'iterative') process²² through which businesses identify, prevent, mitigate and communicate publicly about their actual and potential adverse human rights impacts.²³ Due diligence, in this sense, is a process by which businesses "operationalise" the corporate responsibility to respect human rights outlined by Pillar II of the UNGPs. The UNGPs highlight that the scale and complexity of human rights due diligence is expected to vary according to the size, sector, operational context, ownership and structure of a business, and with the severity of an enterprise's potential adverse human rights impacts. The responsibility to respect human rights is a standard of conduct expected of companies, due diligence being the means to meet this responsibility.²⁴ Unlike general corporate "due diligence" as practiced outside the human rights context, human rights due diligence focuses on identifying risks to rights-holders, rather than risks to the business. It is therefore critical that the business engage with a range of stakeholders, including rights-holders, as part of the due diligence process, in order to properly understand and address its human rights impacts. Since 2011, due diligence as understood under the UNGPs has been incorporated into the OECD Guidelines for Multinational Enterprises and associated guidance.²⁵
- **Due diligence as a general corporate risk management process**: "Due diligence" also refers to processes undertaken by companies, or by legal or other professionals on their behalf, to identify and manage risks to their business. Due diligence processes in this sense are executed, for instance, by parties to corporate mergers and acquisitions; for compliance monitoring purposes, in areas such as anti-corruption or sanctions; or to assess the risks of specific acts, such as employing particular individuals, making a loan, or investing in a particular sector. Due diligence assessments in this sense may not refer to risks to human rights at all. For some audiences, particularly in a business context, "human rights due diligence", as applied in the UNGPs context, may be mistakenly understood to be the same as or similar to the likely more familiar notion of corporate due diligence. However, the two are clearly distinguishable in their legal basis, objectives, scope, character and consequences.
- **Due diligence in international human rights law**: Under human rights treaties, states can have obligations to prevent certain harms to human rights

by private (or 'non-state') actors. Failing to fulfil such 'positive' duties can result in a default by the state on its international legal obligations, where additional conditions are met.²⁶ Certain human rights bodies have expressed the state's positive obligations using the language of "due diligence".²⁷ In this setting, "due diligence" is an obligation of *conduct* expected of states, rather than an obligation of *result*; the standard of conduct to which a state will be held has been expressed, in the broader setting of public international law, as that of "responsible government".²⁸

- **Corporate human rights due diligence as a standard of care:** Some commentators have suggested that, mirroring the concept of due diligence in international human rights law just described, corporate human rights due diligence described by the UNGPs embodies a substantive legal "standard of care" owed by businesses to potential victims of harm resulting from their acts or omissions, or those of their business partners, throughout the value chain of the business in question.²⁹ On this account, a business should be liable for any harms it has caused to human rights itself or via business partners, unless it can prove it performed an adequate due diligence process.³⁰ There are conceptual and legal differences between due diligence obligations on states under human rights treaties and the responsibility of companies to undertake due diligence, despite a common terminology. On the other hand, this does not exclude that states, individually or collectively, may adopt legislation providing for corporate liability (civil or otherwise) for harms linked to due diligence failures.
- **Due diligence as a defence to liability or 'safe harbour':** In areas beyond human rights, some laws provide a defence to liability where a company meets certain criteria, which may be set out in a "safe harbour" exception or due diligence defence.³¹ These are considered in more detail in section 6.3 below.

Policy and scholarly commentaries often do not recognise the spectrum of interrelated but distinct meanings associated with the term due diligence. Individual documents may switch between different meanings or formulations in ways that are not readily transparent to legislators or others. ³² Each sense of due diligence noted above will entail different consequences if given effect in civil liability provisions of mandatory due diligence laws. Full legal analysis, explanatory notes and guidance, debate and consultation with stakeholders are therefore essential prerequisites to enacting national or other due diligence laws and in connection with proposals for including civil liability as an element thereof.

4.2. SCOPE OF HUMAN RIGHTS DUE DILIGENCE: KEY PARAMETERS

Equally, in legislating corporate human rights due diligence requirements, it is important to clarify the different parameters of the due diligence process envisaged by the UNGPs, which may be thought of in terms of:

• **Breadth:** the types of human rights impact considered by the due diligence process;

- **Depth:** how far through the supply or value chain a business's due diligence responsibilities extend; and
- **Content:** the due diligence steps required to be taken by a business in accordance with the legal duty.

Besides these dimensions of human rights due diligence, a statutory due diligence duty can also vary, amongst others, in terms of:

- the class of companies addressed;³³ and
- transparency measures: for instance, requirements on companies to report on their due diligence process or outcomes, or to respond to information requests from third parties.

Each of these elements will influence the potential extent of companies' civil liability for human rights harms, both within a given legislative scheme, where this is provided for, and beyond it, in general tort law and potentially other areas.

4.2.1. Breadth

According to the UNGPs, human rights due diligence should address all "internationally-recognised" human rights.³⁴ When undertaking human rights due diligence, companies should consider risks to all rights-holders affected or potentially affected by the activities of a company or its business partners, rather than selected sub-groups, or only rights-holders affected by specific types of human rights abuses (e.g. human trafficking, forced or child labour). Some due diligence laws do however focus on specific abuses or groups as seen, for instance, in laws on modern slavery³⁵ or child labour.³⁶

French³⁷ and German³⁸ due diligence laws, as well as the European Parliament's proposal³⁹ for EU-level due diligence legislation (the **EP Proposal**), extend the scope of due diligence to certain environmental harms as well as human rights. Although combined environmental and human rights due diligence legislation is not as such contemplated by the UNGPs or currently required by human rights treaties, states' duties under human rights treaties may require the adequate regulation and availability of effective remedies for corporate environmental harms. Defining and assessing "harm" for the purposes of establishing liability under a due diligence law will entail distinct exercises of legal and factual inquiry in relation to environmental damage and human rights abuses.

4.2.2. Depth

Due diligence under the UNGPs considers impacts on human rights caused by a company's own activities but also by its business partners. Companies therefore need to consider risks throughout the value chain to which they are connected via supply, contract, ownership or investment relationships or the use of company products, premises or services.

To align with the UNGPs, due diligence laws may require that companies' due diligence processes evaluate their full value chain. Some laws rely on a full value

chain risk assessment as a trigger for further specific requirements on companies to address or manage risks they pose to human rights.⁴⁰ Other laws restrict due diligence duties to a specific tier of the supply chain.⁴¹ While the pros and cons of different approaches in this respect are debated, there is a possibility that an approach delimited by tier may discourage companies from considering risks in tiers outside the legislation's formal requirements but where the risks of abuses may be higher.⁴² Legislators' choices regarding the depth of companies' statutory due diligence duties will significantly condition the design and operation of any civil liability provisions included in due diligence laws, and vice versa.

4.2.3. Content

The UNGPs outline the main steps of a human rights due diligence process. Yet they recognise that due diligence is flexible, context-dependent, dynamic and iterative in character and, to achieve its goals, must remain a constant "work in progress". Consistently with the UNGPs, businesses can perform human rights due diligence in different ways, relying on variable internal structures, methods, information and mechanisms for securing stakeholder input. What in practical terms respecting human rights requires of businesses will depend on the business context; a full due diligence exercise which spans the entirety of the value chain may not be feasible in a given context.⁴³

Due diligence laws also vary in how they describe what they require of businesses. They may outline specific steps a company must satisfy to comply with its duties under the law or refer to more general criteria. While the French Law requires the production of a Vigilance Plan,⁴⁴ Germany's law specifies a number of specific due diligence elements.⁴⁵ The Dutch Child Labour Law requires companies to investigate whether their goods or services have been produced using child labour and devise a plan to prevent it where instances of child labour are identified. The Norwegian Transparency Act requires a certain class of companies to carry out and publish due diligence assessments related to fundamental rights and decent working conditions.⁴⁶ Whereas some advocates have proposed that due diligence laws should express a standard of care or conduct to be achieved by companies rather than the means of due diligence, this approach has not yet been adopted in practice.⁴⁷

4.2.4. Class of businesses

All companies should undertake human rights due diligence, since the corporate responsibility to respect human rights applies to all businesses, regardless of size or other characteristics. Yet legislators may decide to focus statutory due diligence duties on a subset of businesses for various reasons. Where they do so, the class of companies subject to a statutory due diligence duty may be defined with reference to a range of company characteristics. The scope of the French Law, for example, is defined with reference to a combination of factors, specifically, the number of employees and annual revenue.⁴⁸ As regards SMEs, some proposed laws are restricted to those operating in high-risk sectors or that are publicly listed.⁴⁹ Alternatively, statutory duties may apply to all companies in the first instance, with the identification of risks by companies triggering additional due diligence requirements,

as in the Dutch Child Labour Due Diligence Law.⁵⁰ However the class of companies subject to due diligence is defined, the choices made by legislators on this point will be in interplay with decisions on whether to include civil liability provisions and if so, in what terms.

4.2.5. Transparency

Transparency by companies about human rights risks and measures they have taken to address them supports respect for human rights and can help facilitate remediation. The UNGPs provide that businesses should communicate about their due diligence processes publicly.⁵¹ Mandatory human rights due diligence laws include a range of transparency requirements. The Vigilance Plan required by the French Law, for instance, requires companies to outline what they have done to identify risks and prevent serious violations of human rights and fundamental freedoms, and serious harms to health and safety and the environment. Transparency requirements in other measures are more extensive. Norway's law, for example, requires companies to provide information on how they address human rights and decent working conditions in response to requests from any member of the public.⁵² Under the French Law, complying companies are required to publish Vigilance Plans in line with criteria set down in the law. In the absence of formal guidance on how companies may satisfy the disclosure requirement, the scope of this disclosure obligation has been contested in litigation.⁵³

The inclusion of civil liability provisions in due diligence laws may entail greater anxiety, and opposition, to such laws' transparency provisions. Hence, legislators will need to strike a balance, taking into account the potential value of both such elements within a mandatory due diligence regime.

KEY POINTS:

- "Due diligence" has a range of established meanings which need to be distinguished in discussing and enacting due diligence laws.
- "Due diligence" may be used to refer to: human rights due diligence as described by the UNGPs; a general corporate risk management process; an aspect of states' duties under international laws; a defence to liability or 'safe harbour'; or a standard of care owed by businesses to affected stakeholders.
- How human rights due diligence is defined in mandatory due diligence laws has implications for any associated civil liability regime and potentially liabilities beyond the due diligence law in question.
- Legislators and stakeholders should reflect carefully on how the different dimensions of corporate human rights due diligence (breadth, depth, content), as well as provisions defining the class of businesses subject to due diligence, and transparency requirements, interplay with civil liability provisions when preparing to enact due diligence laws.

GENERAL ELEMENTS OF TORT LIABILITY AND THEIR APPLICATION TO CLAIMS LINKED TO HUMAN RIGHTS ABUSES BY COMPANIES

To succeed in tort proceedings, typically a claimant must show: that a defendant owed her a duty of care; that she suffered a recognised form of harm; that the harm was foreseeable; that the harm was caused by the acts or omissions of the defendant; that such acts or omissions owed to fault on the part of the defendant; and that the imposition of liability is in all the circumstances reasonable.⁵⁴ This section considers how these elements may apply in the context of tort claims established by mandatory corporate human rights due diligence laws.

5.1. ACTIONABLE HARM

As a general principle, a person seeking a civil remedy for a human rights harm will only be able to do so via a domestic court if the law of the particular jurisdiction recognises such a harm as a cause of action, for example, by characterising the harm as a tort or providing a mechanism for a statutory claim. This requires defining an actionable harm. In some countries, actionable harm extends broadly to include emotional, economic, or reputational injuries, as well as violations of privacy, property, or constitutional rights. Torts may include workplace injury; certain forms of environmental damage, occupiers' liability; defamation; intentional torts such as assault, battery, false imprisonment, intentional infliction of emotional distress, and fraud. Still, not all harms suffered by victims, even if they result from a defendant's wrongful acts or omissions, are actionable in civil law. Multiple specific rules, furthermore, govern each such type of actionable harm, again varying by jurisdiction. In the context of personal injury,⁵⁵ for instance, injuries that are "harmless" or "de minimis", are unlikely to be recoverable: "material" harm is generally required.

As noted earlier, violations or abuses of human rights under international law as such are not in general currently recognised as a cause of actionable harm in tort law between private parties. There are exceptions to this general proposition, such as the US Alien Tort Statue which provides a civil cause of action for certain defined breaches of international law,⁵⁶ or recent case law from the Canadian context⁵⁷ which recognises the possibility for a civil cause of action based on breaches of customary international law. Such causes of action can be challenging. For example, It is not exactly clear, for corporate abuses, which country's law is applicable in assessing whether a harm suffered by a claimant also comprised a violation of human rights, or what legal analysis should be applied such a determination. Existing due diligence laws, and jurisprudence on transnational torts, do not provide specific or generally applicable orientation on such matters.

In terms of civil liability provisions included in due diligence legislation, a legislature could seek to define a breach of international human rights standards as an

actionable harm, which was the approach taken in the recently rejected Swiss Responsible Business Initiative proposal (the **Swiss RBI Proposal**).⁵⁸ A less challenging pathway, given the complexities alluded to above, may be to define actionable harm by reference to a breach of an obligation contained in a due diligence law, such as a harm caused by a failure to undertake due diligence in accordance with the terms of the law. This is indeed the approach adopted by the French law. Such a device could work in concert with other pre-existing mechanisms, such as pathways to liability through ordinary tort claims, breaches of consumer law or constitutional or other claims, depending on the particular legal context.

5.2. DUTY OF CARE

In common law systems, before a tort claim can succeed, it must be established that the defendant owed a "duty of care" to the injured party, in the specific circumstances in which the harm occurred.⁵⁹ In common law, a court will determine whether such a duty of care linked the claimant and defendant, taking into account various factors.⁶⁰

Where there is a duty of care, it requires the defendant to exercise a "reasonable" or "responsible" standard of care.⁶¹ One way of understanding civil liability provisions in due diligence laws, then, is that they seek to establish a duty of care between a company and potential victims of human rights abuses linked to the activities of the company or its business partners, as such a wide-ranging duty of care might not otherwise exist.⁶²

Recent due diligence laws rely on different formulations and approaches in establishing a duty of care. The Swiss RBI proposal,⁶³ for instance, would have imposed a duty of care on businesses to "respect internationally recognised human rights and environmental standards"⁶⁴ and to "ensure that human rights and environmental standards are also respected by businesses under their control".⁶⁵

The French Law, by contrast, declines to express a broad corporate duty to respect human rights. Rather, it imposes a specific duty on businesses to publish and implement a "vigilance plan" and permits a claim by any victim of "serious violations of human rights and fundamental freedoms, risks and serious harms to health and safety and the environment", if the victim can show that the company's failure to publish and implement such a plan caused the harm suffered.⁶⁶

The German law requires companies to identify, prevent and mitigate risks in their own operations and arising in their first tier suppliers, and to undertake a risk analysis and develop appropriate preventative measures where the company has "substantiated knowledge" of a violation of human rights or environmental standards in suppliers beyond the first tier.

Obviously, the wider the scope of the duty of care expressed by a due diligence law, the greater the number of potential victims and claimants who may in principle benefit. Yet it will remain for courts to interpret and apply such provisions in individual cases coherently with general norms and elements of civil liability in other areas. For companies with transnational activities and value chains, those who may be potentially affected by their activities or via those of their business partners are dispersed worldwide. Their number, location or identity may be constantly changing. Notwithstanding a broadly framed duty of care in a due diligence law, such considerations, it should be recognised, may dissuade courts from finding the imposition of such a duty of care "reasonable in all the circumstances", or that harm sustained was "foreseeable". In this respect, restricting the duty of care in due diligence laws, for the purposes of civil liability, to certain categories of business activity or tiers of the value chain could carry some potential advantages. Finally, the scope of the due diligence exercise required under the UNGPs will not in general map reliably to the scope of the duty of care in civil liability. Administrative supervision and sanctions can play an important role to bridge the gap in circumstances where the scope of the due diligence requirement is broader than the scope of potential liability.

5.3. STANDARD OF CARE

Besides showing that a defendant owed her a duty of care, in tort a claimant must also establish that her treatment by the defendant fell below the expected "reasonable" or "responsible" standard of care. In litigation, this element comprises two parts: determining the standard of care owed (a question of law); and demonstrating that the defendant's conduct in fact fell below the standard of care owed (a question of fact). In tort proceedings beyond the due diligence scenario, diverse factors may be deemed relevant in determining the latter question, according to the context at hand. The process of human rights due diligence is not disconnected from its outcomes, which can be relevant in assessing the adequacy or reasonableness of human rights due diligence in a particular instance.⁶⁷

Due diligence laws may indicate some elements of the required standard of care. The French law for instance specifies certain elements of the due diligence "vigilance plan" required of companies. If these elements are not delivered by a company, it suggests the required standard of care has not been met. In addition, however, it can be expected that, besides such elements as are specified in due diligence laws, and the UNGPs themselves, formal and informal guidance⁶⁸ on how companies should conduct due diligence, as well as companies' own policy documents and public communications, will be used by courts and others to assess whether a company conducted itself "reasonably" or "responsibly" in a given case.

This will be particularly so where due diligence laws rely on open formulations such as "reasonable due diligence" or "appropriate due diligence"⁶⁹ in defining the required standard of care. This issue will assume special importance where conducting adequate due diligence is deemed by a due diligence law as a defence to corporate liability.⁷⁰

Supply chain or other industry sustainability schemes are also potentially relevant in connection with the standard of care. It might be considered that a company's participation in such schemes, particularly those certified or otherwise approved by regulators,⁷¹ should be deemed probative of meeting the required standard of care in relation to civil liability under a due diligence law. On the other hand, the adequacy of such schemes in addressing the human rights impacts of participating companies is uncertain.⁷² On this basis, the evidentiary value of participation in industry- or supply-chain schemes with regard to assessing whether a company has met the required standard of care (and also in relation to "safe harbour", see Section 6.3 below) seems unclear.

5.4. CAUSATION

In tort law, a defendant's actions or omissions must have *caused* the harm sustained by the claimant for her claim to succeed. Outside of law, factual causation can be loosely assessed by a "but-for" test: would the harm have occurred butfor the defendant's wrongful act or omission? "Legal causation" embodies a similar requirement, but in combination with other elements, including that the harm complained of must have owed to the defendant's wrongful conduct and "foreseeability". Civil law also generally recognises contributory causation, so that damages, if awarded, may be apportioned between multiple defendants each of whom have contributed to injury in a given situation, for instance, with damages being weighted according to their respective roles in causing harm.

Causation in the context of human rights abuses linked to businesses may be amenable to such analysis. In other scenarios, it is harder to see how courts might proceed in determining causation and allocating liability, for instance, where a company has conducted itself in accordance with national legislation; where government bodies have permitted a specific business activity; in situations where a company is in a government joint venture; or where deliberate efforts have been taken by third parties to conceal unlawful conduct from a defendant company.

Due diligence laws vary in their approach to causation. Under the French law, causation is to be determined in line with general principles of French tort law.⁷³ The EP legislative proposal provides that a company may be liable for harms it causes or "contributes to". While a definition of causation is not provided, the EP legislative proposal defines "contribute to" as meaning that "an undertaking's activities, in combination with the activities of other entities, cause an impact, or that the activities of the undertaking cause, facilitate or incentivise another entity to cause an adverse impact." The contribution, it is further stated, must be "substantial", meaning that minor or trivial contributions are excluded. The EP legislative proposal further suggests that "contribution" can be assessed with reference to three factors: (a) the extent to which an undertaking's activities may encourage or motivate an adverse impact by another entity; (b) the extent to which an undertaking could or should have known about the adverse impact or potential for adverse impact, i.e. the degree of foreseeability; and (c) the degree to which any of the undertaking's activities actually mitigated the adverse impact or decreased the risk of the impact occurring.⁷⁴ Hence, on the EP's approach, determining contribution would entail the undertaking of a complex legal and factual analysis.

As noted earlier, a company's human rights due diligence process, in line with the UNGPs, should encompass adverse impacts in all three categories of involvement: "causing", "contributing" or being "directly linked" to human rights impacts.⁷⁵ It has been acknowledged that these three "categories" however function more as a spectrum.⁷⁶ To complicate matters further, evidently there is no clear or automatic relationship between the three categories of involvement declined by the UNGPs and the element of causation required to be made out for the purposes of establishing civil liability.

Pending the application of civil liability provisions of due diligence laws in the courts, it will remain ambiguous whether certain actions or omissions by companies constitute causation as a required element of legal liability. It is unknown at present whether such ambiguity, linked to the prospect of civil liability, will encourage businesses to redouble their due diligence efforts, with the aim of managing human rights risks effectively; or, on the other hand, whether it will discourage due diligence because companies or their legal advisors fear the due diligence exercise will yield a paper trail that may be helpful to claimants in the context of civil liability mechanism.

Specifying the elements of the required due diligence process to be undertaken by companies, either in legislation, or formal guidance, is therefore important.

5.5. FORESEEABILITY

Foreseeability, in the ordinary law of tort, has the function of setting an outer limit to the scope of parties' legal liability. Even if a defendant caused harm to a claimant to whom it owed a duty of care, through actions or omissions falling below the required standard of care, the claim will not succeed unless the court holds that the harm was also reasonably foreseeable, in other words, that it could reasonably have been expected.⁷⁷

It seems likely that mandatory due diligence laws will have some influence on determinations of foreseeability in relevant cases. For instance, where harms complained of by a claimant ought to have been identifiable by a due diligence exercise conducted to a reasonable standard, and a defendant is under a specific legal duty to conduct such due diligence, it would appear harder to argue that the harm caused was unforeseeable. Vice versa, the foreseeability requirement may constrain the scope of liability *prima facie* provided for under due diligence laws, if courts deem harms complained of as unforeseeable.⁷⁸

The UNGPs require that human rights due diligence addresses actual and potential human rights impacts across a company's value chain. They also envisage that a business enterprise may cause or contribute to adverse human rights impacts that were not foreseeable, even with the "best policies and practices".⁷⁹ Again, this highlights discrepancies between the concept of due diligence under the UNGPs and the elements of civil liability. In seeking to enact civil liability provisions in due

diligence laws, legislators and others should be aware of this and the boundaries on liability that may potentially be introduced via foreseeability or similar requirements.⁸⁰

KEY POINTS:

- Not all corporate human rights abuses can be readily analysed as torts.
- The standard elements of torts (including actionable harm, duty of care, standard of care, causation, foreseeability) will generally be required to be met for companies to be held liable under civil liability provisions of due diligence laws.
- Performing due diligence under the UNGPs is not fully equivalent to fulfilment of a duty of care: depending on the exact circumstances to hand, a company that undertakes an adequate due diligence exercise may still be liable for harm; on the other hand, a company that does not undertake adequate due diligence and which causes harm, may not be liable, for other reasons .
- Legislators should be aware of the constraints that will likely be imposed by general tort law principles on the scope of liability under due diligence laws.
- Legislators will need to balance the advantages civil liability provisions against their potential influence on the performance of due diligence by companies covered by mandatory due diligence laws and beyond.

6. FURTHER ISSUES

6.1. STRICT LIABILITY

Generally, as seen above, tort liability follows a finding that a defendant was at fault. Exceptionally, however, the law may establish "strict liability" offences where no fault on the part of a defendant is needed. Such offences, which are narrowly defined or limited in advance to specific factual circumstances, are found in the areas of product liability, defamation and infringement of intellectual property rights, for example.

Some strict liability regimes incorporate a due diligence defence, whereby a defendant can avoid liability by showing it took adequate measures to avoid the harm complained of.⁸¹ The Swiss RBI proposal would have introduced a due diligence defence to strict liability of a controlling company for harm caused by entities under its control.⁸² Other due diligence laws have not however relied on such a mechanism.

In principle, mandatory due diligence laws could combine both fault-based and strict liability. For example, strict liability might be attached to grave harms, such as child labour, forced labour or discrimination, while fault-based liability could attach to other harms. Yet such an approach could also divert due diligence efforts to such harms, even where this was not warranted, in terms of risk. Such factors will need to be weighed carefully in the design of due diligence legislation.

6.2. LIABILITY ACROSS THE SUPPLY OR VALUE CHAIN

The corporate responsibility to respect human rights expressed by the UN Framework and UNGPs extends to all a businesses' activities and relationships, regardless of where they occur in its value or supply chains. This is important because a business' impacts may not be confined to its own activities of those of its immediate suppliers. In some cases, its greatest and most severe impacts may occur at some remove, for instance, in the context of primary production of commodities or "raw materials".

For a variety of reasons, holding parent companies liable for subsidiaries and "lead companies" liable for abuses at the "bottom" of a supply chain may be legally challenging. The immediate perpetrators of human rights abuses may be separated from parent or "lead companies" by many layers of contracting, ownership and geographical distance, for instance. In such a scenario, defendants are likely to argue that the requirements of causation and foreseeability, amongst others, are not met. Indeed, this approach has been taken by defendants even where harms occurred in the first tier of their supply chain.⁸³

Due diligence duties, for reasons explained above, should in themselves make it harder for companies to sustain such arguments, as do supply chain transparency rules. Yet it may be thought that civil liability provisions in due diligence laws should make explicit their extension beyond the first tier, further to avoid this outcome. To this end the French Law provides specifically that a company may be liable for harms where it "exercised control" over or had an "established commercial relationship" with the party immediately responsible for causing harm.⁸⁴ Control, under the French Commercial Code, is defined as "exclusive control" which gives a company "decision-making power, in particular over the financial and operational policies of another entity".⁸⁵ This can be of a legal, de facto or contractual nature and includes first- and lower tier subsidiaries.⁸⁶ Under the French law, companies may hence be liable for harms perpetrated by controlled entities or suppliers in an established commercial relationship where such harms could have been avoided by establishing or implementing a vigilance plan.

Under the EP Proposal, companies would be required to provide remediation for harms they, or undertakings under their control, have caused or contributed to.⁸⁷ Here the concept of control is defined as "the possibility for an undertaking to exercise decisive influence on another undertaking, in particular by ownership or the right to use all or part of the assets of the latter, or by rights or contracts or any other means, having regard to all factual considerations, which confer decisive influence on the composition, voting or decisions of the decision making bodies of an undertaking".⁸⁸

The concept of control in the Swiss RBI proposal included both the control that a parent company may exercise over its subsidiaries but also included "economic control", a sufficiently expansive concept to include the market position of a company in relation to its supplier and the terms of the contract between the two entities.⁸⁹ Accordingly, this approach would have embraced situations where a lead company exercised control over a supplier.⁹⁰ Yet this approach might also entail of discouraging parent companies or lead firms from involving themselves in the management subsidiaries or suppliers.⁹¹

In the UK, rather than control per se, recent case law has focused on the extent to which the parent "voluntarily assumed" or did take over or share with the subsidiary the management of the relevant activity, which "may or may not be demonstrated by the parent controlling the subsidiary". ⁹² A parent may incur responsibility to third parties if, in its published materials, it holds itself out as exercising that degree of supervision and control of its subsidiaries, whether or not it does in fact do so.⁹³ Enactment of statutory duties explicitly linking suppliers and subsidiaries with parent and lead companies may not in all jurisdictions be required, if tort law has evolved to sustain such linkages by other devices, such as expanding the potential scope of a general tortious duty of care.

6.3. DUE DILIGENCE DEFENCE AND "SAFE HARBOUR"

The analysis above has shown that, while the performance of due diligence by a company may be relevant to a determination of civil liability in various ways, it is unlikely to be conclusive of it. This accords with the UNGPs observation that while "conducting appropriate human rights due diligence should help business enterprises address the risk of legal claims against them", on the other hand, "business enterprises conducting such due diligence should not assume that, by itself, this will automatically and fully absolve them from liability for causing or contributing to human rights abuses."⁹⁴

In the context of statutory due diligence schemes, it may be considered that fulfilment of a statutory due diligence requirement should be allowed to operate as a defence, i.e. a company would avoid liability if it could demonstrate it had undertaken a due diligence exercise that was adequate and appropriate in the circumstances. For example, the UK Bribery Act 2010 establishes a criminal offence of failure to prevent bribery.⁹⁵ However, it also establishes a defence for a company that has put in place "adequate procedures" to protect against bribery and corruption.⁹⁶

The EP's legislative proposal, would also provide a form of due diligence defence, allowing "undertakings that prove[d] that they took all due care in line with this Directive to avoid the harm in question, or that the harm would have occurred even if all due care had been taken" should not be held liable for that harm.⁹⁷

Under the Swiss RBI proposal, to avoid liability a company would have needed prove that it exercised appropriate due diligence to prevent the damage in question.⁹⁸

The availability of a human rights due diligence defence to civil liabilities under a due diligence law, or generally, could incentivise businesses to implement human rights due diligence measures, furthering prevention.⁹⁹ Such a preventive role presumes that companies' due diligence exercises are assessed against an appropriate benchmark: cosmetic or "box-ticking" due diligence exercises should not serve to shield companies that have caused actionable harm. This highlights the potential role of formal due diligence guidance, associated with due diligence laws, without which courts may be uncertain of what standard of due diligence to apply. Yet on the other hand, the context-dependent character of due diligence entails that courts would still need to interpret how formal guidance, if produced, should be implemented in a cases' specific circumstances.

Notably, a "safe harbour" doctrine, though similar, is distinguishable from a defence in legal terms. A "safe harbour" may exist where a legislature has intended that a statutory duty be accompanied by a shield from liability in some form. For example, litigation based on disclosures made under the California Transparency in Supply Chains Act 2010 has held that the Act created a "safe harbour" whereby a company would be shielded from civil liability where they truthfully and accurately complied with the requirements of the Act.¹⁰⁰

KEY POINTS:

Fault-based and strict liability

- Strict liability offences are narrowly defined
- Strict liability offences and fault-based liability may be combined in due diligence regimes
- Differentiated approaches may help focus company attention on the gravest abuses but at the same time may divert risk management efforts from more salient risks.

Liability across the supply chain

- There are potential advantages and disadvantages of defining liability provisions with reference to tiers of the supply or value chain that need to be carefully evaluated
- Due diligence laws have applied a range of approaches in defining the extent of companies' potential liability across the supply chain, with reference to control or assumption of responsibility, amongst others

Due diligence as a defence

- A due diligence defence might incentivise businesses to implement human rights due diligence measures, furthering prevention
- However this presumes that companies' due diligence exercises can be assessed in court against appropriate benchmarks and the need for authoritative due diligence guidance

7. CONCLUSION

Due diligence legislation has an important role to play in advancing implementation of the state duty to protect, the corporate responsibility to respect and the right of victims to remediation, as well as more specific elements of the UNGPs and responsible business conduct.

Provisions holding companies accountable for due diligence failures, including by attaching liability and penalties, are key to prevention of abuses and promoting effective risk management by companies.

At the same time, law-makers and others should reflect carefully on the range of mechanisms that can be relied on to achieve such accountability. As illustrated by this briefing linking corporate human rights due diligence as envisaged by the UNGPs to tort liability in mandatory corporate due diligence laws presents complex challenges. Although it is one route to remedy, tort-based liability does not offer a universally appropriate or complete remedy for corporate human rights abuses, while judicial proceedings, given associated cost and delays, will in many cases remain a remedy of last resort for victims.¹⁰¹

Accordingly, any mandatory human rights due diligence measure must be supported by a range of other mechanisms, including enforcement by an empowered and adequately resourced regulator¹⁰² and effective operational level grievance mechanisms.

Finally, while the building blocks of liability mechanisms as highlighted here are common to many jurisdictions, they vary in the exact character according to local context and there is no 'one size fits all' template for the design of human rights due diligence legislation. Accordingly, besides general analyses such as the present one, jurisdictionally-specific studies are needed to inform national and regional legislative processes and debate amongst stakeholders.

	European parliament proposal	French Duty of Vigilance Law	Swiss Popular Initiative on Responsible Business	German Supply Chains Law	Dutch Child Labour Law	Norwegian Transparency Act
Status	Proposal.	In force.	Defeated in a referendum held on 29 November 2020.	Adopted but not in force until 1 January 2023.	Adopted but not in force until 2022.	Adopted but not yet in force. Effective date to be determined.
Scope	Large companies, listed SMEs and SMEs in high risk sectors, domiciled or delivering products or services in the EU.	 Companies incorporated or registered in France for two consecutive fiscal years, and which: employ at least 5,000 people themselves and through their French subsidiaries, or employ at least 10,000 people themselves and through their subsidiaries located in France and abroad. 	All companies that have their registered office, central administration, or principal place of business in Switzerland.	Companies that have their central administration, headquarters or registered office in Germany, provided they have more than 3,000 employees in Germany.	Companies registered in the Netherlands and companies from abroad selling goods and delivering services to Dutch customers.	 All larger companies domiciled in Norway, as well as foreign companies selling products and services in Norway. The law uses the Norwegian Accounting Act to define company size. Companies meeting at least two out of the following three criteria are covered by the act: At least 50 man- years; Turnover of at least 70 million NOK; Balance of at least 35 million NOK
Due Diligence obligation	Requires that Member states lay down rules to ensure that companies carry out a due diligence process aimed at identifying, ceasing, preventing, mitigating, monitoring, disclosing, accounting for, addressing, and remediating the risks	Companies must develop, disclose and implement a "vigilance plan" (plan de vigilance). This plan should include "reasonable vigilance measures adequately to identify risks and prevent serious violations of human	Companies would have been required to carry out appropriate due diligence. In particular they must: identify real and potential impacts on internationally recognized human rights and the environment; take appropriate measures	Companies must introduce iterative and ongoing due diligence processes on the human rights and environmental protections specified in the law. This includes: • Establishing a risk management system	Companies are required to conduct due diligence related to child labour and submit a statement to a supervising authority declaring that they have investigated risks of child labour in their activities and supply chains.	Companies are required to implement due diligence with respect to human rights and decent work, and to document how they work to prevent or limit these risks. The law requires companies to provide or cooperate to ensure remedy where

	European parliament proposal	French Duty of Vigilance Law	Swiss Popular Initiative on Responsible Business	German Supply Chains Law	Dutch Child Labour Law	Norwegian Transparency Act
	with respect to human rights, environmental and governance risks.	rights and fundamental freedoms, risks and serious harms to health and safety and the environment".	to prevent the violation of internationally recognized human rights and international environmental standards, cease existing violations, and account for the actions taken.	 Defining internal responsibilities Carrying out regular risk assessments Putting an internal complaints procedure in place Documenting the fulfillment of the due diligence obligations 		this is due.
Scope of due diligence	The company's own operations and across the whole value chain.	 The company's own operations and business relationships: directly/indirectly controlled companies; subcontractors and suppliers with an "established commercial relationship". 	The company's own operations and controlled companies as well as to all business relationships	The company's own operations including subsidiaries where the parent company exercises determinative influence over the subsidiary, and first tier suppliers. Companies are also required to identify risks linked to suppliers beyond Tier 1 on an ad hoc basis where they obtain "substantiated knowledge" of a potential violation.	The company's own operations and supply chain.	The company's own operations and across the whole value chain.

	European parliament proposal	French Duty of Vigilance Law	Swiss Popular Initiative on Responsible Business	German Supply Chains Law	Dutch Child Labour Law	Norwegian Transparency Act
Transparency	Publish a due diligence strategy on the company's website and upload it on EU platform, and inform stakeholders including workers' representatives, unions, business partners.	Publish a Vigilance Plan in the annual report		Publish annual reports on the fulfilment of the due diligence obligations company's website and submit them to the competent authority.	Statements prepared under the Act have to be lodged within six months from the date the Act enters into force, will be then made publicly available by the national competent authority. Companies are required to produce the statement only once, rather than annually.	Companies must also report on due diligence policies and routines for handling risks to human rights and decent work, including cases of severe risk or harmful incidents to authorities. The information must also be made readily available digitally on the company's websites. The law also provides anyone who so pleases the right to request information from a company on how they manage their due diligence.
Civil liability	Member States are required to establish a liability regime under which undertakings can be held liable in accordance with national law and provide remediation for any harm arising out of potential or actual adverse impacts on human rights, the environment or good	A company may be liable under the French Civil Code if its failure to establish, implement and publish a Vigilance Plan caused harm to fundamental freedoms, health and safety or the environment.	The proposal included a liability mechanism which would have created an avenue for claimants to bring claims in Swiss courts with respect to human rights violations and environmental pollution. The liability mechanism extended to acts of subsidiaries and for economically	No new civil liability mechanism.	No new civil liability mechanism.	No new civil liability mechanism.

European parliament proposal	French Duty of Vigilance Law	Swiss Popular Initiative on Responsible Business	German Supply Chains Law	Dutch Child Labour Law	Norwegian Transparency Act
governance that they, or undertakings under their control, have caused or contributed to by acts or omissions. Liability under this proposal provides a form of due diligence defence as follows: "undertakings that prove that they took all due care in line with this Directive to avoid the harm in question, or that the harm would have occurred even if all due care had been taken" should not be held liable for that harm.		controlled companies. A potential shield from liability was available if the company could show that it had undertaken appropriate due diligence.			

ENDNOTES

- 1 UN Office of the High Commissioner for Human Rights, "Guiding Principles on Business and Human Rights: Implementing the 'Protect, Respect and Remedy' Framework" (HR/PUB/11/04, 2011), available at <<u>https://www.ohchr.org/documents/publications/</u> <u>GuidingprinciplesBusinesshr_eN.pdf</u>>("UNGPs").
- 2 GP 23, UNGPs.
- 3 GP 1, UNGPs.
- 4 For example, only 14.6% of 1000 companies surveyed in 2019 reported on actual human rights impacts and only 3.6% explained the outcomes of the management of those risks according to the Alliance for Corporate Transparency's 2019 Research Report https://www.eciia.eu/2020/05/ alliance-for-corporate-transparency-2019-research-report/>. See also the findings of the study conducted by BIICL et al, for DG JUST, Study on due diligence requirements through the supply chain, European Commission (January 2020) which found that voluntary initiatives, even when backed by transparency do not sufficiently incentivise good practice; there exists wide stakeholder support, including from frontrunner businesses, for mandatory EU due diligence; and that 70% of businesses responding to the survey conducted for the study agreed that EU regulation might provide benefits for business, including legal certainty, level playing field and protection in case of litigation. Recent studies have also revealed shortcomings in the implementation of the UNGPs by business including the work of the Corporate Human Rights Benchmark < https://www.worldbenchmarkingalliance.org/publication/ <u>chrb/</u>>, as well as studies commissioned by the German government, Final Report of the NAP monitoring process 2018-2020 (2020) <https:// www.auswaertiges-amt.de/de/aussenpolitik/themen/aussenwirtschaft/ wirtschaft-und-menschenrechte/monitoring-nap/2131054>, the Dutch government, Evaluation and revision of policy on Responsible Business Conduct (2020) < https://www.government.nl/topics/responsiblebusiness-conduct-rbc/evaluation-and-renewal-of-rbc-policy>, the Danish Institute for Human Rights, Documenting business respect for human rights: A snapshot of Danish companies (2020) <https://www. humanrights.dk/news/danish-companies-fail-document-their-workhuman-rights>, and Trinity College Dublin Irish Business and Human Rights: A snapshot of large firms operating in Ireland (2020) <https:// www.tcd.ie/business/assets/pdf/CSI-BHR-2020-Report.pdf>, which show a low uptake of human rights due diligence processes by companies when done on a voluntary basis.
- 5 ECCJ, "Map: Corporate accountability legislative progress in Europe" (14 June 2021) <<u>https://corporatejustice.org/publications/map-corporate-</u>

accountability-legislative-progress-in-europe/>.

- 6 As the UNGPs indicate, "The responsibility of business enterprises to respect human rights is distinct from issues of legal liability and enforcement, which remain defined largely by national law provisions in relevant jurisdictions" Commentary to GP 12, UNGPs.
- 7 This is referred to in the UNGPs as a "smart mix" of measures, meaning a combination of national and international, mandatory and voluntary measures to foster business respect for human rights. Commentary to GP3, UNGPs.
- 8 These include measures enacted in: the Netherlands, Kamerstukken I, 2016/17, 34 506, (Dutch Child Labour Due Diligence Law); Germany Gesetz über die unternehmerischen Sorgfaltspflichten in Lieferketten (16 July 2021) (German Law) <<u>https://www.bgbl.de/xaver/bgbl/start.xav?startbk=Bundesanzeiger_BGBl&jumpTo=bgbl121s2959.pdf#_bgbl_%2F%2F*%5B%40attr_id%3D%27bgbl121s2959.pdf%27%5D_1633956617769>; Norway, Prop 150 L (2020–2021) (Norwegian Transparency Act) <<u>https://www.regjeringen.no/no/dokumenter/prop.-150-l-20202021/id2843171/?ch=1</u>>; and Switzerland, Code des Obligations: Contre-projet indirect à l'initiative populaire "Entreprises responsables pour protéger l'être humain et l'environnement" (19 June 2020) <<u>https://www.parlament.ch/centers/eparl/curia/2016/20160077/Texte%20pour%20le%20vote%20final%20 2%20NS%20F.pdf</u>>.</u>
- 9 See also the work of the OHCHR Accountability and Remedy Project, <<u>https://www.ohchr.org/EN/Issues/Business/Pages/</u> <u>OHCHRaccountabilityandremedyproject.aspx</u>> in particular the Report on the relationship between human rights due diligence and determinations of corporate liability (1 June 2018) A/HRC/38/20/Add.2.
- 10 *ICCPR* 2(3). See generally, C. Methven O'Brien, *Business and Human Rights: A Handbook for Legal Practitioners* (2018, Council of Europe), Chapter 3 Access to Remedy. 6 *ICERD*, Article 14 Convention Against Torture and Other Cruel, Inhuman or Degrading Treatment or Punishment (CAT) and Article 39 CRC. The *ICESCR* and the *ICEDAW* do not explicitly provide for a right of remedy. It may be argued, however, that a right to remedy is implicit in these instruments since human rights treaties presume national implementation and require this for the effectiveness of the rights they articulate, or alternatively based on a norm of customary international law to provide a remedy for human rights violations. Moreover, optional Protocols to both *ICESCR* and *ICEDAW* which are now in force permit rights-holders to bring complaints before these instruments' respective treaty bodies. See also *UDHR* Art 8.
- 11 See European Court on Human Rights Factsheet: Extra-territorial Jurisdiction of States Parties to the European Convention on Human Rights, (July 2018) and American Commission on Human Rights Business

and Human Rights: Inter-American Standards, (2019) at 90-101; C. Methven O'Brien, "The Home State Duty to Regulate the Human Rights Impacts of TNCs Abroad: A rebuttal", *Business and Human Rights Journal*, *3(1) (2018), 47-73.*

- 12 Common law and civil law systems differ in how they formulate and refer to the law of non-contractual obligations. This paper will refer to general principles of tort or its equivalents, such as delict or civil law actions, but the precise application of these principles will depend on the national laws of the particular jurisdiction.
- 13 See also the US *Trafficking Victims Prevention Act of 2000* which allows victims of trafficking to seek compensation from companies via civil claims in US courts. In Yem *Ban, Sophea Bun, Sem Kosal, Nol Nakry, Keo Ratha, Sok Sang and Phan Sophea v Doe Corporations, Phatthana Seafood Co Ltd, Rubicon Resources LLC, SS Frozen Food Co Ltd and Wales and Co Universe Ltd* No 2:16-cv-04271, (CD Cal, 15 June 2016) Cambodian villagers who worked in Thailand producing shrimp for export to the US relied on this mechanism.
- 14 Typically, where this is allowed, the law relies on the "identification doctrine" by which a company may be criminally liable if it can be shown that senior officers who could be regarded as the "directing mind and will" of the company had the requisite intent, see eg: *Tesco Supermarkets Ltd v Nattrass* [1972] AC 153; *R v St Regis Paper Co Ltd* [2012] 1 Cr App R 14. The US takes a far more extensive approach, whereby a company can be liable if one of its agents commits a crime which benefits the company, regardless of their place in the corporate hierarchy, under the doctrine of *respondeat superior*, ie liability does not require commission of an act by a senior official.
- 15 See for example, Part 2.5 of the Australian *Criminal Code Act* (Cth) 1995 under which a company can be liable if it can be taken to have authorised or permitted the commission of an offence by having in place a corporate culture which either tolerated non-compliance or failed to promote compliance.
- 16 Usually in cases where there is no intent element to the offence. The offence of failure to prevent bribery in section 7 of the UK *Bribery Act 2010* operates on a similar basis.
- 17 See for example, *Nerijus Antuzis & ors v DJ Houghton Catching Services Ltd & ors* [2019] EWHC 843 (QB) where company directors were held liable for inducing breach of contract and failing to act in the best interests of the company by engaging in wage theft practices.
- 18 There are exceptions to this general rule, conferred by statute, see for example in the UK *Contracts (Rights of Third Parties) Act 1999.* In some jurisdictions a third party beneficiary can enforce a contract (for example Australia, see *Trident General Insurance Co Ltd v. McNiece Bros Pty Ltd* (1988) 165 CLR 107). Other exceptions may arise in certain circumstances where there is a collateral contract, or involvement of an agent.

- 19 Recent jurisprudence has acknowledged the limited utility of clauses requiring parties to adhere to certain human rights related standards, such as those concerning health an in breachd safety. In a recent example from the UK, a suit was brought against a UK company that had acted as the agent in the sale of a ship that was decommissioned in a shipyard in Bangladesh. The claim alleged that the UK company owed a duty of care to a worker who died in the course of his work owing to poor health and safety conditions in the shipyard. The Court of Appeal of England and Wales noted that the relevant contract included a standard clause requiring the buyer of the ship to sell only to a breakers yard which had good health and safety practices, but that the unhappy reality was that the seller would have no interest in enforcing such a provision, and therefore the buyer had no incentive to comply with the clause: *Begum v Maran (UK) Ltd* [2021] EWCA Civ 326 at [68]–[71].
- 20 For example, a suite of lawsuits brought against Samsung in France which alleged that labour abuses were present in its factories including instances of child labour, despite Samsung having a "zero tolerance" policy toward child labour. See "Samsung lawsuit (re misleading advertising & labour rights abuses)" (*Business & Human Rights Resource Centre*) <<u>https://www.business-humanrights.org/en/latest-news/samsung-lawsuit-re-misleading-advertising-labour-rights-abuses/>.</u>
- 21 A number of US claims have used this formulation: see Barber v Nestlé USA Inc No 8:2015cv01364 (CD Cal, 14 December 2015); McCoy v Nestlé USA No 3:2015cv04451 (ND Cal, 29 March 2016); Wirth v Mars Inc No 8:2015cv01470 (CD Cal, 5 February 2016); Hodson v Mars Inc No 4:2015cv04450 (ND Cal, 17 February 2016); Dana v The Hershey Company No 3:2015cv04453 (ND Cal, 29 March 2016); Sud v Costco Wholesale Corporation No 3:2015cv03783 (ND Cal, 24 January 2016); and Tomasella v Nestle USA Inc Case No 1:18-cv-10269, US District Court for the District of Massachusetts.
- 22 Due diligence has also been described as a "bundle of interrelated processes": *Corporate human rights due diligence emerging practices, challenges and ways forward* (16 July 2018) UN Doc A/73/163, para 10.
- 23 UNGP 17.
- 24 Rachel Davis, Legislating for Human Rights Due Diligence: How Outcomes for People Connect to the Standard of Conduct (August 2021) <<u>https://</u> <u>shiftproject.org/hrdd-outcomes-standard/</u>>; OHCHR, The Corporate Responsibility to Respect Human Rights: An Interpretive Guide (2012) <<u>https://ohchr.org/Documents/Publications/HR.PUB.12.2_En.pdf</u>>, p6
- 25 OECD Guidelines for Multinational Enterprises (OECD 2011); OECD Due Diligence Guidance for Responsible Business Conduct" (OECD 2018).
- 26 C Methven O'Brien, Business and Human Rights: A Handbook for Legal Practitioners, pp.19-22.
- 27 E.g. Human Rights Committee (HRC), "General Comment 31: Nature of the General Legal Obligation Imposed on States Parties to the Covenant"

(29 March 2004) UN Doc CCPR/C/21/Rev1/Add13, para 8.

- 28 T Koivurova, Due Diligence, Oxford Encyclopaedia of Public International Law
- 29 J Bonnitcha and R McCorquodale, "The Concept of 'Due Diligence' in the UN Guiding Principles on Business and Human Rights" (2017) 28 *European Journal of International Law* 899, 900; cf. John Ruggie and John Sherman, "The Concept of 'Due Diligence' in the UN Guiding Principles on Business and Human Rights: A Reply to Jonathan Bonnitcha and Robert McCorquodale" 28(3) *European Journal of International Law* 92.1
- 30 L Smit et al, *Study on due diligence requirements through the supply chain: FINAL REPORT* (European Commission 2020) 262: "The occurrence of a harm suggests that the company would be liable, unless it could show that it has done everything that could have been reasonably expected in the circumstances".
- 31 Note, the notions of due diligence defence and safe harbour exception, though similar, may be legally distinct. To avail of a due diligence defence, a company might argue that it conducted due diligence to the standard required. By contrast, a safe harbour exception might set out criteria that, if met, could exclude a company's potential liability and a claimant's cause of action L Smit and C Bright *The concept of a 'safe harbour' and mandatory human rights due diligence* (CEDIS Working Paper No 1, December 2020).
- 32 These include "reasonable" due diligence, "appropriate" due diligence and "adequate" measures. A study for Amnesty International highlighted confusion between the terms "HRDD", "duty of care" and related terms from non-common law jurisdictions, with refence to the French "devoir de vigilance" and German "menschenrechtliche Sorgfaltspflicht". In particular, the term "vigilance" used in the French Law "has much more specific, codified conditions than the very general duty of care": A Rühmkorf and L Walker, "Assessment of the concept of 'duty of care' in European legal systems for Amnesty International" (European Institutions Office, September 2018) 5.
- 33 Whether public entities, commercial or otherwise, should have statutory human rights due diligence duties is an important question however not one considered in this briefing.
- 34 GP 12, UNGPs.
- 35 UK Modern Slavery Act 2015 and Australian Modern Slavery Act 2018.
- 36 Dutch Child Labour Due Diligence Law.
- 37 French Law.
- 38 German Law.
- 39 European Parliament resolution of 10 March 2021 with recommendations to the Commission on corporate due diligence and corporate accountability (2020/2129(INL) (10 March 2021) <<u>https://www.europarl.</u> <u>europa.eu/doceo/document/TA-9-2021-0073_EN.html</u>>
- 40 See for example the Dutch Child Labour Due Diligence Law and the

Norwegian Transparency Act.

- 41 German Law.
- 42 See e.g. Ruggie's letter to members of the German cabinet, dated 9 March 2021: <<u>https://shiftproject.org/wp-content/uploads/2021/03/Shift_John-Ruggie_Letter_German-DD.pdf</u>>.
- 43 Commentary GP17, UNGPs.
- 44 For example, the requirements for a Vigilance Plan in the French Law.
- 45 DEUTSCHER BUNDESTAG, 'Gesetzesentwurf der Bundesregierung: Entwurf eines Gesetzes über die Unternehmerischen Sorgfaltspflichten in Lieferketten', Drucksache 19/28649, 19.04.2021, available at <u>https://dserver.bundestag.de/btd/19/286/1928649.pdf</u>; See further: BUNDESMINISTERIUM FÜR ARBEIT UND SOZIALES, 'Sorgfaltspflichtengesetz - Gesetz über die unternehmerischen Sorgfaltspflichten zur Vermeidung von Menschenrechtsverletzungen in Lieferketten', Website, 2021, available at <u>https://www.bmas.de/DE/</u> <u>Service/Gesetze-und-Gesetzesvorhaben/gesetz-unternehmerische-</u> <u>sorgfaltspflichten-lieferketten.html.</u>
- 46 It should be noted that neither the Dutch Child Labour Due Diligence Law nor the Norwegian Transparency Act incorporate a civil liability mechanism.
- 47 This approach, however, has been proposed. For example, in the recently defeated legislative proposal from the Swiss Responsible Business Initiative (**Swiss RBI**).
- 48 French Law.
- 49 For example, the Swiss RBI would have only required SMEs in particularly risky sectors to comply. A similar approach was takin in the EP Proposal whereby listed and "high risk" SMEs would have been required to comply.
- 50 Dutch Child Labour Due Diligence Law.
- 51 GP21; Commentary to GP21, UNGPs.
- 52 Norwegian Transparency Act. The UK and Australian Modern Slavery Acts require the production of reports on the management of modern slavery risks in the supply chain, without mandating any particular steps that the company should take to do so: UK *Modern Slavery Act 2015* and Australian *Modern Slavery Act 2018*
- 53 See for example two recent pieces of litigation against Total concerning the adequacy of its vigilance plans, one brought by 14 French local authorities and five French NGOs alleging that Total had not sufficiently identified the human rights and environmental risks linked to its contribution to climate change, summary available at the Business and Human Rights resource Centre <<u>https://www.business-humanrights.org/</u> <u>en/latest-news/total-lawsuit-re-climate-change-france/</u>>, and another brought by French and Ugandan civil society organizations alleging that Total had not adequately addressed risks concerning its operations in Uganda, summary available at the Business and Human Rights Resource Centre: <<u>https://www.business-humanrights.org/en/latest-news/totallawsuit-re-failure-to-respect-french-duty-of-vigilance-law-in-operations-</u>

2021

in-uganda/>

- 54 In cases of "strict liability" a defendant may be liable in the absence of fault; in addition, common law and civil jurisdictions may differ in these elements.
- 55 Personal injury, as a class of tort law, may refer to disease or an impairment of a person's physical or mental condition
- 56 See the US Alien Tort Statute, which grants US federal courts jurisdiction over "any civil action by an alien for a tort only, committed in violation of the law of nations or a treaty of the United States." 28 U.S.C. § 1350. Notably, however, the extraterritorial application of this law and the amenability of corporations to be sued under its provisions have been brought into question by recent cases such as *Kiobel v. Royal Dutch Petroleum Co* 133 S. Ct. 1659 (2013) and *Jesner v. Arab Bank PLC* 138 S. Ct. 1386 (2018).
- 57 Nevsun Resources Ltd. v. Araya [2018] SCCA No. 26.
- 58 The Swiss RBI proposal envisaged liability flowing as follows: "Companies are also liable for damage caused by companies under their control where they have, in the course of business, committed violations of internationally recognized human rights" Swiss Coalition for Corporate Justice ("SCCJ"), "The Initiative Text with Explanations" <<u>https://</u> corporatejustice.ch/wp-content/uploads//2018/06/KVI_Factsheet_5_E. pdf>.
- 59 See e.g. in English law: *Donoghue v Stevenson*: A duty to "take reasonable care to avoid acts or omissions which you can reasonably foresee would be likely to injure your neighbour...[i.e.] persons who are so closely and directly affected by my act that I ought reasonably to have them in contemplation as being so affected when I am directing my mind to the acts or omissions which are called into question" [1932] AC 562, 580.
- 60 In this context, common law court will balance foreseeability against "the cost of avoiding the harm, and the benefits to society foregone if the activity in question is not carried on" Deakin, Johnston and Markesinis, *Markesinis and Deakin's Tort Law*, Oxford University Press, 5th edition, 2003 p80. Notably, in English law, determining whether the damage lies within the scope of the duty of care assumed by the defendant has superceded individual consideration of issues of causation and remoteness, see e.g. *Manchester Building Society v Grant Thornton UK LLP* [2021] UKSC 20 and *Khan v Meadows* [2021] UKSC 21.
- 61 Notably, the term "duty of care" does not always find a precise equivalent in civil law systems. In addition, the term "duty of care" can also carry different meanings within common law systems. Besides indicating an element of tort claims, for instance, "duty of care" is also used in corporate governance to refer to directors' fiduciary duties. A recent study for Amnesty International found that there was considerable confusion between the terms "HRDD", "duty of care" and quasi-equivalent terms from non-common law jurisdictions such as the French "devoir de

vigilance" and German "menschenrechtliche Sorgfaltspflicht". See A Rühmkorf and L Walker, "Assessment of the concept of 'duty of care' in European legal systems for Amnesty International" (European Institutions Office, September 2018).

- 62 Cf. Heron II [1969] 1 AC 350: the defendant will be liable for any type of damage which is reasonably foreseeable as liable to happen even in the most unusual case unless the risk is so small that a reasonable man would in the whole circumstances feel justified in neglecting it.
- 63 Swiss Coalition for Corporate Justice ("SCCJ"), "The Initiative Text with Explanations" <<u>https://corporatejustice.ch/wp-content/</u> <u>uploads//2018/06/KVI_Factsheet_5_E.pdf</u>>.
- 64 See proposed Article 101a §2a put forward by the Swiss Responsible Business Initiative.
- 65 See proposed Article 101a §2a put forward by the Swiss Responsible Business Initiative. In effect this would have made a business covered by the law vicariously liable for harms caused by companies it controlled.
- 66 "any person found to have breached the obligations defined in article L. 225-102-4 of this Code may be held liable and required to repair the damage that would have been avoided had he/she complied with said obligations" (art L 225-102-5).
- 67 Rachel Davis, Legislating for Human Rights Due Diligence: How Outcomes for People Connect to the Standard of Conduct (August 2021) <<u>https://</u> <u>shiftproject.org/hrdd-outcomes-standard/</u>>
- 68 For example, due diligence guidance published by the OECD: OECD Due Diligence Guidance for Responsible Business Conduct (2018); OECD Due Diligence Guidance for Responsible Supply Chain of Minerals from Conflict-Affected Areas, 2016; OECD-FAO Guidance for Responsible Agricultural Supply Chains, 2016; OECD Due Diligence Guidance for Responsible Supply Chains in the Garment and Footwear Sector, 2017; OECD Responsible Business Conduct for Institutional Investors: Key Considerations for due diligence under the OECD Guidelines for MNEs, 2016.
- 69 Swiss Coalition for Corporate Justice ("SCCJ"), "The Initiative Text with Explanations" <<u>https://corporatejustice.ch/wp-content/</u> <u>uploads//2018/06/KVI_Factsheet_5_E.pdf</u>>.
- 70 The Swiss RBI proposal envisaged a due diligence defence which avoided liability "if they can prove that they took all due care ... to avoid a loss or damage, or that the damage would have occurred even if all due care had been taken". See the Swiss Coalition for Corporate Justice ("SCCJ"), "The Initiative Text with Explanations" <<u>https://corporatejustice.ch/wp-content/uploads//2018/06/KVI_Factsheet_5_E.pdf</u>>.
- 71 See for example Regulation (EU) 2017/821 of the European Parliament and of the Council of 17 May 2017 laying down supply chain due diligence obligations for Union importers of tin, tantalum and tungsten, their ores,

and gold originating from conflict-affected and high-risk areas.

- 72 See for example GIZ et al, Human rights in the palm oil sector: The responsibility of purchasing companies: Limits and potentials of certification (2020) <<u>https://www.forumpalmoel.org/imglib/Studien/</u><u>Human%20Rights%20Study.pdf</u>>
- 73 Causation can be determined in two ways: adequate causality [causalité adéquate] which seeks to find the most likely determining cause of the damage, or the equivalence of conditions [équivalence des conditions] which is based on the idea that each factor contributed to cause the damage. See S Brabant and E Savourey, "France's Corporate Duty of Vigilance Law: A closer look at the penalties faced by companies" (2017) *Revue Internationale de la Compliance et de L'Éthique des Affaires Supplément à La Semaine Juridique Entreprise et Affaires N° 50 3* <<u>https://media.business-humanrights.org/media/documents/d32b6e38d5c199f8912367a5a0a6137f49d21d91.pdf</u>>.
- 74 EP Proposal, art 3. The draft law can be found in the Annex to European Parliament resolution of 10 March 2021 with recommendations to the Commission on corporate due diligence and corporate accountability (2020/2129(INL)) <<u>https://www.europarl.europa.eu/doceo/document/</u> <u>TA-9-2021-0073_EN.pdf</u>>.
- 75 Under the UNGPs, a company's responsibility to remediate abuses extends where it causes or contributes to abuses; its responsibility is rather to "exercise leverage" in the case of direct linkage.
- 76 J Ruggie, "Comments on Thun Group of Banks Discussion Paper on the Implications of UN Guiding Principles 13 & 17 in a Corporate and Investment Banking Context" (Harvard Kennedy School, 21 February 2017).
- 77 The devices of "foreseeability", "remoteness" and "proximity" have had similar functions.
- 78 Recent cases are testing the boundaries of foreseeability. In the shipbreaking case referred to above, the claimant alleges that a UK agent in the sale of a vessel should have known or foreseen that the low sale price of the vessel meant that it would be ultimately sent to a shipyard in Bangladesh at a yard with negligible safety standards. While the case involved a challenging fact pattern, the Court of Appeal recently decided to allow it to proceed in the English courts. *Begum v Maran (UK) Ltd* [2021] EWCA Civ 326 at [7].
- 79 Commentary GP22. in this scenario, the UNGPs expect that a business will provide for or cooperate in remediation.
- 80 Such as "fair, just and reasonable in all the circumstances".
- 81 See UK *Bribery Act 2010*, s7. see also Decreto Legislativo 8 giugno 2001, n 231, Disciplina della responsabilità amministrativa delle persone giuridiche, della società e delle associazioni anche prive di personalità giuridica, a norma dell'articolo 11 della legge 29 settembre 2000, n

300. See Fédération International pour les droits humains (FIDH) et al, "Italian Legislative Decree No. 231/2001: A Model for Mandatory Human Rights Due Diligence Legislations?" (November 2019) <<u>https://</u> e6e968f2-1ede-4808-acd7-cc626067cbc4.filesusr.com/ugd/6c779a_ d800c52c15444d74a4ee398a3472f64c.pdf>. See further G LeBaron and A Rühmkorf, "Steering CSR Through Home Art Regulation: A Comparison of the Impact of the UK Bribery Act and Modern Slavery Act on Global Supply Chain Governance" (2017) 8 *Global Policy* 15; I Pietropaoli et al, *A UK Failure to Prevent Mechanism for Corporate Human Rights Harms* (BIICL, 11 February 2020) 48–55.

- 82 See proposal for art 101a(2)(c) of the Swiss Constitution which would have provided for specific liability of a controlling company for the harm caused by a controlled company: "companies are also liable for damage caused by companies under their control ... They are not liable however if they can prove that they took all due care ... to avoid the damage, or that the damage would have occurred even if all due care had been taken".
- 83 In another recent case from Germany where it was alleged that a German company should owe a duty of care towards employees of a foreign supplier and was therefore liable in negligence for failing to ensure adequate fire safety precautions in the factory Jabir v KiK Textilen und Non-Food GmbH 7 O 95/15. The duty was alleged to be based on the fact that the German company purchased around 75% of the textiles manufactured by the factory and intervened in the factory's operations, including by incorporating a code of conduct in its supplier contracts entitling it to monitor safe working conditions. C Terwindt et al, "Supply chain liability: Pushing the boundaries of the common law?" (2017) 8 Journal of European Tort Law 261 <<u>http://repository.essex.</u> ac.uk/20684/1/Supply%20chain%20liability-%20pushing%20the%20 boundaries%20of%20the%20common%20law%20-%20pre%20 edit%20version.pdf>
- 84 Recent cases in the UK have held that liability may attach where a company assumes responsibility by taking an active role in the management of another entity or represents that they exercise a degree of supervision, even where they do not do so in fact. Although these cases concerned liability of a parent over a subsidiary, there is nothing in the decisions which suggests that these principles could not be applied to the relationship between a lead firm and a supplier *Okpabi and ors v Royal Dutch Shell Plc and anor* [2021] UKSC 3; *Vedanta Resources Plc anor v Lungowe and ors* [2019] UKSC 20.
- 85 S Brabant, C Michon and E Savourey, "The Vigilance Plan: Cornerstone of the Law on the Corporate Duty of Vigilance" (2001) 50 *Revue Internationale de la Compliance et de l'Éthique des Affaires* 93.
- 86 See N Bueno and C Bright, "Implementing Human Rights Due Diligence Through Corporate Civil Liability" (2020) 69 International & Comparative Law Quarterly 789 <<u>https://papers.ssrn.com/sol3/papers.cfm?abstract</u>

<u>id=3689241</u>>.

- 87 EP Proposal, art 19(2). The draft law can be found in the Annex to European Parliament resolution of 10 March 2021 with recommendations to the Commission on corporate due diligence and corporate accountability (2020/2129(INL)) <<u>https://www.europarl.europa.eu/</u> doceo/document/TA-9-2021-0073_EN.pdf>.
- 88 EP Proposal, art 3. The draft law can be found in the Annex to European Parliament resolution of 10 March 2021 with recommendations to the Commission on corporate due diligence and corporate accountability (2020/2129(INL)) <<u>https://www.europarl.europa.eu/doceo/document/</u> <u>TA-9-2021-0073_EN.pdf</u>>.
- 89 Swiss Coalition for Corporate Justice, "Rapport explicatif de l'initiative populaire fédérale «Entreprises responsables : pour protéger l'être humain et l'environnement»" 43 <<u>https://initiative-multinationales.ch/wpcontent/uploads//2018/05/20170912_Erl%C3%A4uterungen-FR.pdf</u>>.
- 90 See N Bueno and C Bright, "Implementing Human Rights Due Diligence Through Corporate Civil Liability" (2020) 69 International & Comparative Law Quarterly 789 <<u>https://papers.ssrn.com/sol3/papers.cfm?abstract_id=3689241>.</u>
- 91 G Holly, "Zambian farmers can take Vedanta to court over water pollution. What are the legal implications?" (*Business & Human Rights Resource Centre*, 12 April 2019) <<u>https://www.business-humanrights.org/en/latest-news/zambian-farmers-can-take-vedanta-to-court-over-water-pollution-what-are-the-legal-implications-2</u>/>; M Croser et al, "Vedanta v Lungowe and Kiobel v Shell: The Implications for Parent Company Accountability" (2020) 5 *Business and Human Rights Journal* 130.
- 92 Okpabi and ors v Royal Dutch Shell Plc and anor [2021] UKSC 3 at [147]. control should be "just a starting point" See also Vedanta Resources Plc anor v Lungowe and ors [2019] UKSC 20.
- 93 Okpabi and ors v Royal Dutch Shell Plc and anor [2021] UKSC 3 at [148].
- 94 Commentary to GP17.
- 95 See s7 *UK Bribery Act 2010.* This provides that a relevant commercial organisation is guilty of an offence if a person associated with such organisation bribes another person intending to obtain or retain: business for such organisation; or an advantage in the conduct of business for such organisation. A person guilty of an offence under section 7 is liable on conviction on indictment to a fine.
- 96 Some have advocated for a similar model to be adopted in the context of human rights harms. See Irene Pietropaoli et al, A UK Failure to Prevent Mechanism for Corporate Human Rights Harms, BIICL, (February 2020) <<u>https://www.biicl.org/documents/84_failure_to_prevent_final_10_feb.</u> pdf>
- 97 EP Proposal, art 19. The draft law can be found in the Annex to European Parliament resolution of 10 March 2021 with recommendations to the

Commission on corporate due diligence and corporate accountability (2020/2129(INL)) <<u>https://www.europarl.europa.eu/doceo/document/</u> TA-9-2021-0073_EN.pdf>.

- 98 Proposal art 101(2)(c) Swiss Constitution.
- 99 See OHCHR, "Improving accountability and access to remedy for victims of business-related human rights abuse: the relevance of human rights due diligence to determinations of corporate liability" (1 June 2018) UN Doc A/HRC/38/20/Add2, para 29 <<u>https://undocs.org/A/HRC/38/20/Add.2</u>>.
- 100 Barber v Nestlé USA Inc No. 8:2015cv01364 (C.D. Cal. December 14, 2015) at p787.
- 101 EU Agency for Fundamental Rights, *Business and Human Rights Access to Remedy* (2020) <<u>https://fra.europa.eu/sites/default/files/fra_uploads/fra-2020-business-human-rights_en.pdf</u>>; C Bright et al, *Access to legal remedies for victims of corporate human rights abuses in third countries* (European Parliament Subcommittee on Human Rights 2019) <<u>https://www.europarl.europa.eu/RegData/etudes/STUD/2019/603475/EXPO_STU(2019)603475_EN.pdf</u>>.
- 102 C Methven O'Brien and O Martin-Ortega, EU Human Rights Due Diligence Legislation: Monitoring, Enforcement and Access to Justice for Victims (European Parliament Subcommittee on Human Rights 2020).

